Author's response to reviews

Title: How accessible and acceptable are self-management support interventions for men with long term conditions? A systematic review and meta-synthesis of qualitative studies

Authors:

Paul M Galdas (paul.galdas@york.ac.uk)
Zoe Darwin (z.j.darwin@leeds.ac.uk)
Lisa Kidd (lisa.kidd@gcu.ac.uk)
Christian Blickem (christian.blickem@manchester.ac.uk)
Kerri McPherson (kerri.mcpherson@gcu.ac.uk)
Kate Hunt (kate.hunt@glasgow.ac.uk)
Peter Bower (peter.bower@manchester.ac.uk)
Simon Gilbody (simon.gilbody@york.ac.uk)
Gerry Richardson (gerry.richardson@york.ac.uk)

Version: 2
Date: 22 September 2014

Author's response to reviews:

1.) We have changed the manuscript font to Times New Roman
2.) Line Numbering has been added to the main manuscript
3.) As this was a secondary synthesis of data, ethical approvals were not required. We have added this statement in the methods section on p.5.
4.) A list of abbreviations used has been provided on p.21
5.) Competing Interests have been declared on p.21
6.) Authors' Contributions have been added on p.21
7.) Acknowledgements have been added on p.21